Author's response to reviews

**Title**: Relationship between Body Mass Index (BMI) and body fat percentage, estimated by bioelectrical impedance, in a group of Sri Lankan adults: a cross sectional study.

**Authors:**

Chathuranga Ranasinghe (chath_r@yahoo.com.au)
Prasanna Gamage (jpgamage@yahoo.com)
Prasad Katulanda (pkatulanda@yahoo.com)
Nalinda Andraweera (andraweera@yahoo.com)
Sithira Thilakarathne (sithiratw@yahoo.com)
Praveen Tharanga (tdpt83@gmail.com)

**Version**: 2 **Date**: 11 January 2013

**Author's response to reviews**: see over
11th January 2013

Victorino Silvestre
Journal Editorial Office
BioMed Central, United Kingdom,

Dear Sir,

Submission of research article to BMC Public Health- Submission of revisions

Title: Relationship of body mass index (BMI) and body fat percentage, estimated by bioelectrical impedance, in a group of Sri Lankan adults: a cross sectional study.

Manuscript ID: 1818735563878771

Thank you very much for your letter in request to submit the changes needed to the above mentioned manuscript.

I will be providing the existing sentence and the replacing sentence/change to the specific place in the article.

(1) Specific name of the Ethics committee that granted approval for the study.

Page 4, paragraph 3, line 8

Existing sentence: ‘Institutional ethical approval was taken to carry out the study.’

Replacing sentence: ‘Ethical approval was taken by Ethics Review Committee, Faculty of Medicine University of Colombo Sri Lanka’.

Further I would like to make following two changes,

(2) Page 4, paragraph 3, line 4

Existing sentence: ‘… an exhibition venue during a medical exhibition conducted by, Faculty of Medicine, University of Colombo Sri Lanka in 2006.’

Change: The year should be changed to 2008 not 2006.

(3) Page 4, paragraph 3, line 7

Existing sentence: ‘Confidentiality was maintained during the storage of data.’

Replacing sentence: ‘Confidentiality was maintained during the storage, retrieval and analysis of data.’
We hope that the above changes would assist your process of revision. Please be kind enough to contact me for any other details. We would further like to extend our regret for the inconvenience caused by the late reply.

Thank you,

Dr. Chathuranga Ranasinghe
Corresponding author
Lecturer, Allied Health Sciences Unit
Faculty of Medicine, University of Colombo
Sri Lanka
Email: chath_r@yahoo.com.au